Author's response to reviews

Title: Breast and spleen metastasis of squamous carcinoma from the uterine cervix: a case report

Authors:

meryem MA aitelhaj (m_meryem_a@hotmail.com)
siham SL lkhovali (s.lkhovali@gmail.com)
anouar AB boukir (boukir@yahoo.fr)
mustapha ME elkabous (e-elkabous@gmail.com)
halima HA abassain (abahssainhalima@yahoo.fr)
hind HM mrabti (h_mrabti@gmail.com)
basma BE elkhanoussi (dr_basma_2@hotmail.com)
hassan HE errihani (h-errihani@yahoo.fr)

Version: 2
Date: 1 April 2014

Author's response to reviews: see over
Dear Editor

Please find enclosed our manuscript “Breast and spleen metastasis of squamous carcinoma from the uterine cervix: a case report” by Meryem Aitelhaj and al; which we would like to submit for a publication in the “medical journal of case reports” for your consideration.

As you know the metastasis to the breast from extramammary malignancies are extremely rare, the uterine cervix origin is reported only in numerous cases in the literature. The isolated spleen metastases of solid tumor are also rare, and uterine cervix origin is reported only in 3 cases. At the best of our knowledge that is the first case of the association of two unusual metastatic sites of uterine cervical carcinoma, the breast and spleen without widespread disease.

We believe our findings would appeal to the readership of “medical journal of case reports”.

Please address all correspondence to: Dr AITELHAJ Meryem.

I confirm that it has not been published and it is not under consideration elsewhere.

All authors have approved the manuscript and agree with its submission to “medical journal of case reports”

Thank you for your time and consideration;

We look forward to hearing from you, at your earliest convenience

Sincerely, yours

Dr Aitelhaj Meryem